Author’s response to reviews

Title: Assessment of Peripheral Blood Lymphocyte Subsets in Children with Nutritional Iron Deficiency anemia (Two Egyptian Centers Study)

Authors:

Sanaa Aly (engylaklok@gmail.com)
Hanan Fayed (hananfayed@yahoo.com)
Ahlam Ismail (ahlam1976@hotmail.com)
Gehan Abdel Hakeem (gehanlotfy72@yahoo.com)

Version: 3 Date: 13 Dec 2017

Author’s response to reviews:

Editor Comments:

1. Textual Overlap:

We note that the current submission contains some textual overlap with other previously published works, in particular:


And


This overlap mainly exists in the Introduction (Paragraph 1 of your manuscript - ii) and Discussion (Throughout - i) sections.
While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please re-phrase these sections to minimize overlap.

REPLY:
Had revised and rephrased

2. Copy-Edit:

Unfortunately, there are some minor language issues throughout the manuscript. Please ensure that you have thoroughly checked your manuscript for any other language errors. We recommend that you ask a native English speaking colleague to help you copy-edit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.

REPLY:
Had revised and corrected

3. Title Page:

Please remove the information of funding, conflicts of interest and authors’ contributions from the title page. This information should only be found in the Declarations section.

REPLY:
Had revised and corrected
4. Abstract:

Please include a list of three to ten keywords or phrases after the abstract.

REPLY:

Had revised and added

5. Manuscript Sections:

a. Please change the ‘Introduction’ heading to ‘Background’

REPLY:

Had revised and corrected

b. Please incorporate ‘Limitations of Study’ into either the discussion or the conclusions, as it currently does not fit into our formatting guidelines.

REPLY:

Had revised and corrected

6. Abbreviations:

Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.
7. Ethics Approval and Consent to Participate:

a. Please change the subheading ‘Approval’ to ‘Ethics Approval and Consent to Participate’.

REPLY:
Had revised and corrected

b. Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

REPLY:
Had revised and edited

8. Consent for Publication:

Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

REPLY:
Had revised and corrected
9. Availability of Data and Materials:

Please indicate where the raw dataset generated and/or analysed in this current study will be available from. If you wish to share the data through correspondence then please include the statement “The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.” If you do not wish to share your data, please state this along with the reason.

REPLY:

Had revised and corrected